Author's response to reviews

Title: Radiological Findings In Patients Undergoing Revision Endoscopic Sinus Surgery: A retrospective case series study

Authors:

Hisham S Khalil (hisham.khalil@phnt.swest.nhs.uk)
Ahmed Z Eweiss (azews@yahoo.com)
Nicholas Clifton (nicholas.clifton@phnt.swest.nhs.uk)

Version: 3 Date: 11 April 2011

Author's response to reviews: see over
Dear editor,

Thank you for considering our article “Radiological Findings In Patients Undergoing Revision Endoscopic Sinus Surgery: A retrospective case series study” for publication. The article was part of a registered audit in our hospital, and thus ethics committee approval was not required as it was a research project. A statement has been added in the methods to highlight this.

Yours sincerely,

Authors of the article.